
CENWP-OP-B                                                                                                                           28 January, 2002

MEMORANDUM FOR CENWP-OP, ATTN: Gary Johnson

SUBJECT: Bonneville Lock and Dam, Fishway and Fish Activities for week 04 of 2002, which covers the period 20 through 26 January 2002.
1.  OPERATION SUMMARY.

1.1 Daily river flow averaged 117.0 to 173.3kcfs.  Daily project tailwater elevation averaged 12.8’ to 16.0’ msl.  Daily project forebay elevation averaged 72.0’ to 75.2’ msl.  River temperature averaged 46° F.  Daily turbidity averaged 5.0 to 7’ on the secchi disk.
1.2 Spill Bays 18 and 1 are closed for spill deflector work.  
1.3 Unit 5 remains out of service for turbine rehab.  Unit 17 remains out of service for VBS slot modification work.  Unit 4 went out of service from 0500 on 20 January until 1512 on 21 January and from 0700 until 0936 on 23 January for Bank ¾ work.  Unit 13 went out of service from 1318 until 1624 on 20 January due to a cooler water leak.   Unit 3 returned to service for its 72 hour testing on 1606 on 22 January.
1.4 Second Powerhouse fish units provide attraction flow for the Washington Shore fish ladder.  The units are placed on standby from approximately 0000-0300 nightly, if drawdown exceeds 1.0’.  F1 remains out of service for winter maintenance.  As coordinated with NMFS, F2 is supplying all attraction flow at this time. (see Misc. 5.4)
2.  ACTIVITIES. 
2.1 AWS Closures. 

All Bradford Island AWS valves are closed for winter maintenance.

22 January- FV5-9 closed for cleaning from 0706 until 0712

22 January- FV6-9 closed for cleaning from 0718 until 0746

22 January- FV5-3, FV5-4 closed for cleaning from 0820 until 1415

23 January- FV6-9 closed for cleaning from 1540 until 1600

26 January- FV5-9 closed for cleaning from 1457 until 1513

2.2 STS/VBS Inspections.  All STSs have been removed for winter maintenance.

2.3 Dewatering and Fish Salvage.  No salvage or dewatering activities this reporting week.

3.  RESEARCH.

3.1 WDFW Fish Counting.  24-hour video count ongoing at the WA Shore ladder.

3.2 PSMFC Adult PIT.  Installation underway at Bradford Island ladder.

3.3 WDFW Winter Steelhead tagging.  Trapping continues three or four days a week.

3.4 NMFS Lamprey.  Tracking continues.  Plans for a trap in FV3-9 AWS channel underway.  More lamprey plates to be installed in the Washington Shore ladder in February.

4.  FISHWAYS. 
4.1 Project biologists inspected 22-24 January this reporting week.

4.2 Fish Passage Plan Observations.

DATE
LOCATion
FPP Deviation
CAUSE
RESPONSE
TR/WAR

All week
PH2 electronic velocity meter.
Meter not yet working
Need new cable.
Will be installed in February.


4.3 Fish Ladder Temperatures.  Temperatures are measured 3 weirs down from the count station at Bradford Island and at the junction of the UMT and the main ladder at Washington shore.  Temperature probes are out of service for calibration.

4.4 Sensor Calibration.  Calibration was not checked as there were no sensors in service this reporting week.

4.5 Zebra Mussel Monitoring.  No signs of colonization were observed this reporting week. 

5. MISCELLANEOUS

5.1 The B-Branch and Cascade Island computers were installed.  Until sensors have been calibrated, entrance gates, diffusers, valves FV4-3, FV4-4, FV5-3, FV5-4 are operated in manual.

5.2 Weir 37 float control doesn’t adequately control water level over the weir.  A sonic transducer will be installed this winter, which should correct the problem.  

5.3 To aid spawning Chum below Bonneville, a minimum tailwater restriction of 11.4’ went into effect at 1054 on 28 November.  The restriction will continue until further notice.

5.4 As coordinated with NMFS, the ITC will not be used for AWS water.  F2 is being operated to maintain a 1.5’ differential.  In an effort to reduce the debris load during daylight hours, the fish unit is put on standby from 1800 until 0500 to float trash.   Plant material is a major debris problem.

5.5 PH2 ITC was opened from 1319 until 1535 on 20 January and from 0800 until 1115 on 22 January to remove debris.

 DEBORAH L. CHENOWETH

                                 

                                                             Operations Project Manager
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